Author's response to reviews

Title: Genital warts and infection with human immunodeficiency virus in high-risk women in Burkina Faso: a longitudinal study

Authors:

Andrea Low (andrea.low@lshtm.ac.uk)
Tim Clayton (tim.clayton@lshtm.ac.uk)
Issouf Konate (kletio@yahoo.fr)
Nicolas Nagot (n-nagot@chu-montpellier.fr)
Abdoulaye Ouedraogo (ouarma74@yahoo.fr)
Charlotte Huet (charlotte_huet@yahoo.fr)
Marie-Noelle Didelot-Rousseau (mndidelot@orange.fr)
Michel Segondy (m-segondy@chu-montpellier.fr)
Philippe van de Perre (p-van_de_perre@chu-montpellier.fr)
Philippe Mayaud (philippe.mayaud@lshtm.ac.uk)

Version: 3 Date: 18 January 2011

Author's response to reviews:

To the editor:

I am submitting our final version of the manuscript, as requested. I am writing to provide you with details of any changes made.

1. All highlights and tracking have been removed.
2. A statement has been added to the beginning of author's contributions, to the effect that all authors have read and approved the final manuscript.
3. I have attempted to add the Yerelon Study Group to the author's list, as written on the manuscript first page, but was unable to do so as there is no email address. I have therefore left it on the manuscript only. I have added the members of the study group to the acknowledgments section.
4. I have reviewed the typography of the manuscript and have made the following changes:
   a. p10 line 4, I have added 2 commas to the multivariate analysis sentence.
   b. p11 line 7, I have added the aHR for the effect of primary education.
   c. p13 line 14- changed is not to has not been
   d. p13 line 17- removed a comma
   e. p13 line 18- changed strong to good.

Thank you very much for accepting this manuscript. I hope that the changes made correspond to your requests.

Sincerely,

Andrea Low